Author’s response to reviews

Title: Radiofrequency Catheter Ablation of Ventricular Arrhythmias Arising from the Region above Pulmonary Valve

Authors:

Jin Li (wyd0115@126.com)
Cheng Zheng (793601136@qq.com)
Zhi-Rui Liu (liuzhirui0109@163.com)
Jun Ma (henrymuch@163.com)
Ge Jin (gejinjnn63@163.com)
Wei-Qian Lin (373806983@qq.com)
Yao-Yao Wang (1365899427@qq.com)
jia-feng Lin (linjiafeng_wzmcfey@163.com)

Version: 3 Date: 29 Sep 2019

Author’s response to reviews:

Dear editor:

Thank you sincerely for spending your precious time in reviewing our manuscript. We have made the revisions based on your requirement.

Sincerely,

Jia-feng Lin.

Response to editor:

1. Role of Funders

Please note the role(s) of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript, which should be declared in the Funding section.

Answer: We have declared the role of the funding body in the Funding section.

2. Formatting
In order to be in line with the submission guidelines for BMC Cardiovascular Disorders research articles (found in detail here: https://bmccardiovascdisord.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article), please amend the following section titles:

-Please rename "Introduction" to "Background"

-Please rename "Materials and Methods" to "Methods"

-Please rename "Conflict of interest" to "Competing interest"

Answer: We have corrected accordingly.

3. Clean manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Answer: We have uploaded the manuscript, tables and figures accordingly.